Author's response to reviews

Title: The effect of health literacy on knowledge and receipt of colorectal cancer screening: a survey study

Authors:

David P Miller Jr (dmiller@wfubmc.edu)
Caroline D Brownlee (cbrownl4jhmi.edu)
Thomas P McCoy (tmccoy@wfubmc.edu)
Michael P Pignone (michael_pignone@med.unc.edu)

Version: 4 Date: 29 March 2007

Author's response to reviews:

March 28, 2007

BioMed Central Editorial Production Team
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK

Dear Editors:

We are delighted that BMC Family Practice has accepted our manuscript for publication. Our manuscript with the requested formatting revisions is attached.

We have made the following changes:

1. Title was changed to sentence case.

2. Academic qualifications have been removed from authors' names.

3. Country has been added for each author's affiliation details.

4. One p-value in the abstract has been changed from <0.001 to <0.01 (line 5 of Results).

5. P-value for race in Table 2 has been rounded to 0.02 (from 0.024) for formatting consistency with the rest of the table.

If you require any other changes, please let us know. It has been a pleasure to work with your editorial staff.

Sincerely,

David P. Miller, Jr., MD
Wake Forest University School of Medicine
Section of General Internal Medicine
Medical Center Boulevard
Winston-Salem, NC 27157
T: 1-336-713-5218
e-mail: dmiller@wfubmc.edu